Author's response to reviews

Title: Identifying the determinants of premature mortality in Russia: overcoming a methodological challenge.

Authors:

Susannah Tomkins (susannah.tomkins@lshtm.ac.uk)
Vladimir Shkolnikov (shkolnikov@demogr.mpg.de)
Andreev Evgeni (andreev@demogr.mpg.de)
Kiryanov Nikolay (kirnik@igma.udm.ru)
Leon A. David (david.leon@lshtm.ac.uk)
McKee Martin (martin.mckee@lshtm.ac.uk)
Lyudmila Saburova (sabur@udm.net)

Version: 2 Date: 14 May 2007

Author's response to reviews:

Please see the attached manuscript which includes an abstract.